Author's response to reviews

Title: A randomised trial to evaluate the immunogenicity, reactogenicity, and safety of the 10-valent pneumococcal non-typeable Haemophilus influenzae protein D conjugate vaccine (PHiD-CV) co-administered with routine childhood vaccines in Singapore and Malaysia

Authors:

Fong Seng Lim (fong_seng_lim@nhgp.com.sg)
Mia Tuang Koh (kohmt@ummc.edu.my)
Kah Kee Tan (kktalk@gmail.com)
Poh Chong Chan (paecpc@nus.edu.sg)
Chia Yin Chong (Chong.Chia.Yin@kkh.com.sg)
Yeo Wee Shung Yehudi (yehudi_yeo@nhgp.com.sg)
Yee Leong Teoh (teohyl@yahoo.com)
Fakrudeen Shafi (fakrudeen.a.shafi@gsk.com)
Marjan Hezareh (marjan.x.hezareh@gsk.com)
Kristien Swinnen (kristien.m.swinnen@gsk.com)
Dorota Borys (dorota.d.borys@gsk.com)

Version: 2
Date: 3 June 2014

Author’s response to reviews: see over
Dear Ms Ramos,

Please find enclosed a re-submission of the manuscript entitled “A randomised trial to evaluate the immunogenicity, reactogenicity and safety of the 10-valent pneumococcal non-typeable Haemophilus influenzae protein D conjugate vaccine (PHiD-CV) co-administered with routine childhood vaccines in Singapore and Malaysia”. We would like to consider it for publication as a research article in the BMC Infectious Diseases.

As advised, this manuscript has been amended to include line and page numbers in the main text file of the manuscript.

Thank you very much for considering our manuscript for publication in the BMC Infectious Diseases.

Yours sincerely,

Dr. Lim Fong Seng,
(On behalf of the authors)
First and Corresponding Author
National Healthcare Group Polyclinics
Clinical Programmes and Management (CPM)
3 Fusionopolis Link, #03-08
Nexus@one-north (South Lobby)
Singapore 138543
Tel. +65 64966914 Fax: +65 64966857. E-mail address: fong_seng_lim@nhgp.com.sg